Author's response to reviews

Title: Epidemiology of complications of male circumcision in Ibadan, Nigeria

Authors:

Linus I Okeke (liokeke@yahoo.com)
Adanze A Asinobi (adasinobi@yahoo.com)
Odunayo S Ikuerowo (kerowq@yahoo.com)

Version: 5 Date: 15 August 2006

Author's response to reviews: see over
We have now included sections on acknowledgement, funding and competing interests. Three of our colleagues have also assisted with copy editing of the manuscript. My first language is English. I did my postgraduate studies at the Institute of Urology, University of London. I hope you will find this version acceptable but I will be quite glad to effect any further corrections that you may find necessary.

COMPETING INTEREST

We declare that we have no competing interests.

AUTHOR CONTRIBUTION

1. LIO: made substantial contributions to conception and design, analysis and interpretation of data; drafting the manuscript, revising it critically for important intellectual content; and gave final approval of the version to be published.

2. AA: made substantial contributions to design, data collection, and manuscript revision for important intellectual content and gave final approval of the version to be published.

3. SOI: made substantial contributions to design, data collection, analysis, manuscript revision for important intellectual content and gave final approval of the version to be published.

FUNDING

No funds were received from any funding body for any part of this study.

ACKNOWLEDGEMENT
We thankfully acknowledge the assistance of Drs A.O. Afolabi, S. Ademola and O.O. Ogundoyin who copy edited the manuscript.